Author's response to reviews

Title: Chest x-Ray findings in late onset congenital diaphragmatic hernia, a rare emergency easily misdiagnosed as hydropneumothorax.

Authors:

Vassil N Zefov (vnzefov@dha.gov.ae)
Maryam A Almatrooshi (sweetmind77@hotmail.com)

Version: 3
Date: 12 October 2015

Author's response to reviews: see over
Dear Editor-in-Chief,

Please find our enclosed revised manuscript, as recommended by referee 2 the manuscript title has been changed to “Chest x-Ray findings in late onset congenital diaphragmatic hernia, a rare emergency easily misdiagnosed as hydropneumothorax” as it would be more representative of the manuscript content. We even added detailed operative findings which was not mentioned previously which we believe have caused confusion on the diagnosis, as the operative findings showed the posterior diaphragm having tiny posterior rim and hypoplastic left lung which goes with the diagnosis of late onset congenital diaphragmatic hernia Please address all correspondence to (Maryam Anas Almatrooshi, Email: sweetmind77@hotmail.com). We look forward to hearing from you at your earliest convenience.

With our best regards,